Author's response to reviews

Title: Characteristics and outcomes of patients with eclampsia and severe pre-eclampsia in a rural hospital in Western Tanzania: a retrospective medical record study

Authors:

Rob Mooij (r.mooij1983@gmail.com)
Joseph Lugumila (lugumila@gmail.com)
Masumbuko Y Mwashambwa (masuyt@yahoo.co.uk)
Ipyana H Mwampagatwa (imwampagatwa@yahoo.com)
Jeroen van Dillen (jeroen.vandillen1@radboudumc.nl)
Jelle Stekelenburg (jelle.stekelenburg@online.nl)

Version: 6
Date: 22 August 2015

Author's response to reviews: see over
To:
Jason Pepito
Editorial Office
BMC pregnancy and childbirth

Re: Cover letter for

“Characteristics and outcomes of patients with eclampsia and severe pre-eclampsia in Ndala Hospital, Tanzania: a retrospective medical record study”
by R Mooij, J Lugumila, MY Mwashambwa, IH Mwampagatwa, J van Dillen and J Stekelenburg

August 22nd, 2015

Dear Sir,

As discussed through email, hereby we would like to submit the 5th revision of the above mentioned paper. We have addressed the comment about guarding the patients’ privacy by removing table 5 completely.

We are happy to address any other issues.

Yours sincerely,

Rob Mooij, Medical Officer Ndala July 2010 - January 2013, r.mooij1983@gmail.com
Joseph Lugumila, Medical Officer in charge September 2010-present

Masumbuko Y Mwashambwa
Ipyana H Mwampagatwa
Jeroen van Dillen
Jelle Stekelenburg